**Author's response to reviews**

**Title:** Learning while evaluating: the use of an electronic evaluation portfolio in a geriatric medicine clerkship

**Authors:**

- Gustavo Duque (gustavo.duque@mcgill.ca)
- Adam Finkelstein (adam.finkelstein@mcgill.ca)
- Ayanna Roberts (ayanna.roberts@mail.mcgill.ca)
- Diana Tabatai (diana.tabatai@mcgill.ca)
- Susan L Gold (susan.gold@mcgill.ca)
- Laura Winer (laura.winer@mcgill.ca)

**Version:** 5  **Date:** 22 December 2005

**Author's response to reviews:**

Dear Mr. Austin. We have made the changes according to the publisher's recommendations. The changes include:

- Title and abstract are now identical in the manuscript and in the submission system.
- The list of authors has been written exactly as they are in the submission system
- The title is now sentence case
- Figure titles have been removed.

Thank you for your suggested changes. We hope that the online production process will go go straightforward.

Sincerely,

Gustavo Duque, MD,PhD